Author's response to reviews

Title: Recall of intestinal helminthiasis by HIV-infected South Africans and avoidance of possible misinterpretation of egg excretion in worm/HIV co-infection analyses

Authors:

Vera VJ Adams (vera.arendse@mrc.ac.za)
Miles MB Markus (medsynth@yahoo.co.uk)
Zilungile ZL Kwitshana (zilungile.kwitshana@mrc.ac.za)
Muhammad MA Dhansay (ali.dhansay@mrc.ac.za)
Lize L van der Merwe (lize.van.der.merwe@mrc.ac.za)
Gerhard G Walzl (gwalzl@sun.ac.za)
John JE Fincham (john.fincham@mrc.ac.za)

Version: 6 Date: 12 May 2006

Author's response to reviews:

Changes specified by Tony Austin have been made, as well as other improvements. Overall as follows:
1. The title is more concise.
2. The key words have been improved.
3. The abstract is structured to: Background, Methods, Results, Conclusions.
4. Table placement indications have been removed from the text.
5. In ref 31, "in press" has been replaced by publication details.
6. A new reference 47 has been added.
7. "Acknowledgements" have been added.

John Fincham